
Machine status

• Technical stop 1:
• Main intervention for prototype wirescanner.
• Restart ok after fixing TFB ring 2 and timing problem of distributor ring 1.

• The White Rabbit B Train distribution reliability run re-started.

• New TFB progressing with new HW and FESA class. 
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Beam status
• All operational beams available.

• Beams for LHC special runs (W25/W26) checked:
• Van der Meer on all 4 rings.

• LHC 100ns.

• LHC 50ns.

• Request to re-check all 4 MD3305 cycles: “BSRT & BGI vs WS cross-
calibration”
• dp/p = 0.5

• Em.long. = 0.3 eVs

• Intensity = 1E11 ppb.

• Em. Trans. = 1,2,3,4
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